Author's response to reviews

Title: In vitro evaluation of antibiotics' combinations for empirical therapy of suspected methicillin resistant Staphylococcus aureus severe respiratory infections

Authors:

Lorenzo Drago (lorenzo.drago@unimi.it)
Elena De Vecchi (elena.devecchi@unimi.it)
Lucia Nicola (microbio@unimi.it)
Maria Rita Gismondo (microbio@unimi.it)

Version: 3 Date: 6 September 2007

Author's response to reviews:

We are submitting a new version of our manuscript revised according to suggestions of the The BioMed Central Editorial Production Team.

Specifically:
All tracking and highlighting have been removed from the manuscript.
All underlining has been removed from the manuscript.
Reference 18 is limited to only one page (page 1).
We tried to reformat table 1 into A4 portrait orientation, but unfortunartely to do so, we should reduce character dimension to 9 or 10. Therefore, we have not modified it. However if you like, we can reduce the character to 9
We have removed the full stop from the title of table 4.
We hope that this new version could be suitable for publication,
Looking forward to hear from you,
Best wishes
Lorenzo Drago